Author's response to reviews

Title: Addressing inequity to achieve the maternal and child health Millennium Development Goals: looking beyond averages

Authors:

George M Ruhago (George.Ruhago@isf.uib.no)
Frida N Ngalesoni (Frida.Ngalesoni@isf.uib.no)
Ole N Norheim (Ole.Norheim@isf.uib.no)

Version: 3 Date: 20 June 2012

Author's response to reviews: see over
Dear Editor in Chief, BMC Public health journal

Subject: SUBMISSION OF A REVISED MANUSCRIPT FOR EVALUATION

I am enclosing herewith a revised manuscript entitled “Addressing inequity to achieve the maternal and child health Millennium Development Goals: looking beyond averages” and a point by point to each change.

Thank you for your time and consideration

Sincerely,

George M. Ruhago
George.Ruhago@isf.uib.no

Response to the editorial team comments

1. Data availability: Please document within the Methods section of your manuscript as to whether the data used for the study is openly available or as to whether you received permission (and by who) to use this.

   The data were extracted from the Tanzania Demographic and Health survey report of 2010. Permission was sought and obtained from the Tanzania’s National Institute of Medical Research (NIMR) as required by Tanzania Laws regulating health related research. Sentence to reflect that has been added on page 5